Author’s response to reviews

Title: Effect of lysine supplementation on Hypertensive Men and Women in selected peri-urban community in Ghana

Authors:

Frederick Vuvor (fredvuvor@yahoo.com)

Husein Mohammed (husein.mohammed@mail.mcgill.ca)

Thomas Ndanu (nutcaresoft@gmail.com)

Obed Harrison (obedharrison@gmail.com)

Version: 3 Date: 18 Jul 2017

Author’s response to reviews:

No response